Author’s response to reviews

Title: Economic burden of symptomatic iron deficiency - A Survey among Swiss women

Authors:

Patricia Blank (patricia.blank@unibas.ch)
Yuki Tomonaga (yuki.tomonaga@uzh.ch)
Thomas Szucs (thomas.szucs@unibas.ch)
Matthias Schwenkglenks (matthias.schwenkglenks@uzh.ch)

Version: 5 Date: 12 Feb 2019

Author’s response to reviews:

Dear Editor

Please find attached the clean version of the manuscript including the Ethical statement.

Sincerely yours,

Patricia Blank